Check for updates

Citation: Guzzetta G, Trentini F, Poletti P, Baldacchino FA, Montarsi F, Capelli G, et al. (2017) Effectiveness and economic assessment of routine larviciding for prevention of chikungunya and dengue in temperate urban settings in Europe. PLoS Negl Trop Dis 11(9): e0005918. https://doi. org/10.1371/journal.pntd.0005918

Editor: Pattamaporn Kittayapong, Faculty of Science, Mahidol University, THAILAND

Received: April 7, 2017

Accepted: August 29, 2017

Published: September 11, 2017

Copyright: © 2017 Guzzetta et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: This study was funded by the Autonomous Province of Trento, Research fund "Grandi Progetti 2012", project LExEM (Laboratory of excellence for epidemiology and modelling) awarded to AR. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript. RESEARCH ARTICLE

Effectiveness and economic assessment of routine larviciding for prevention of chikungunya and dengue in temperate urban settings in Europe

Giorgio Guzzetta¹*, Filippo Trentini², Piero Poletti^{1,2}, Frederic Alexandre Baldacchino³, Fabrizio Montarsi⁴, Gioia Capelli⁴, Annapaola Rizzoli³, Roberto Rosà³, Stefano Merler¹, Alessia Melegaro^{2,5}

1 Fondazione Bruno Kessler, Trento, Italy, 2 Dondena Centre for Research on Social Dynamics and Public Policy, Bocconi University, Milan, Italy, 3 Department of Biodiversity and Molecular Ecology, Fondazione Edmund Mach, San Michele all'Adige (Trento), Italy, 4 Laboratory of Parasitology, Istituto Zooprofilattico Sperimentale delle Venezie, Padova, Italy, 5 Department of Policy Analysis and Public Management, Bocconi University, Milan, Italy

* guzzetta@fbk.eu

Abstract

In the last decades, several European countries where arboviral infections are not endemic have faced outbreaks of diseases such as chikungunya and dengue, initially introduced by infectious travellers from tropical endemic areas and then spread locally via mosquito bites. To keep in check the epidemiological risk, interventions targeted to control vector abundance can be implemented by local authorities. We assessed the epidemiological effectiveness and economic costs and benefits of routine larviciding in European towns with temperate climate, using a mathematical model of Aedes albopictus populations and viral transmission, calibrated on entomological surveillance data collected from ten municipalities in Northern Italy during 2014 and 2015. We found that routine larviciding of public catch basins can limit both the risk of autochthonous transmission and the size of potential epidemics. Ideal larvicide interventions should be timed in such a way to cover the month of July. Optimally timed larviciding can reduce locally transmitted cases of chikungunya by 20% - 33% for a single application (dengue: 18-22%) and up to 43% - 65% if treatment is repeated four times throughout the season (dengue: 31-51%). In larger municipalities (>35,000 inhabitants), the cost of comprehensive larviciding over the whole urban area overcomes potential health benefits related to preventing cases of disease, suggesting the adoption of more localized interventions. Small/medium sized towns with high mosquito abundance will likely have a positive cost-benefit balance. Involvement of private citizens in routine larviciding activities further reduces transmission risks but with disproportionate costs of intervention. International travels and the incidence of mosquitoborne diseases are increasing worldwide, exposing a growing number of European citizens to higher risks of potential outbreaks. Results from this study may support the planning and timing of interventions aimed to reduce the probability of autochthonous transmission as well as the nuisance for local populations living in temperate areas of Europe.

Competing interests: The authors have declared that no competing interests exist.

Author summary

Larvicides are a key tool to prevent the growth of mosquito populations and decrease both the risks of outbreaks of mosquito-borne diseases and the nuisance deriving from bites. In order to assist municipalities from temperate areas in Europe in effectively planning vector control programs, we modelled the effect of larviciding in public areas on populations of *Aedes albopictus* using mosquito collection data from 10 municipalities in Northern Italy, over two years with very different temperature conditions. We then evaluated the resulting probabilities of potential outbreaks of chikungunya and dengue and their expected final sizes, and we compared the intervention costs to the economic and health benefits due to the avoidance of clinical cases. By assessing several different intervention strategies, we found that the optimal timing should be centred on the month of July, corresponding to the period of maximal growth of the mosquito population. Municipality-wide interventions are likely to be cost-effective in small-to-medium towns (below 35,000 inhabitants) even where mosquito infestation is moderate, whereas for larger cities a neighbourhood-based intervention should be considered. The involvement of citizens to apply larvicides within private premises resulted effective but generally too costly.

Introduction

During the last decade, Europe has faced outbreaks of mosquito-borne diseases (MBD) such as dengue and chikungunya, following the continuous importation of human cases in areas with established competent vectors such as the invasive mosquito Aedes (Stegomyia) albopictus (Skuse) [1]. Vector control interventions can be implemented by local authorities to keep in check mosquito abundance and consequently reduce the epidemiological risk. Adulticide spraying rapidly reduces the number of mosquitoes, but its effect is short-lived [2]. For this reason, it is particularly indicated in situations where the transmission risk needs to be reduced drastically and quickly, such as when an individual is diagnosed with an MBD, to prevent or curtail an outbreak [3]. Since the effectiveness of reactive measures decreases with the delay between outbreak initiation and implementation of control [4], a better approach may consist in preventive interventions. Treatment of potential breeding sites with larvicide products has a delayed impact in reducing adult populations [3], but experimental studies show that their effect lasts for several weeks [5], making them better suited for preventive routine control. The main limit to larviciding as a control option is the proportion of breeding sites that are actually accessible to interventions by public health authorities. To overcome this limit, education campaigns may be carried forward to encourage citizens to remove and treat potential breeding sites from their private premises during the mosquito season [6, 7]. Mathematical modelling of MBD associated with cost-effectiveness analyses can help optimizing routine vector control interventions [8] with respect to constraints in human and financial resources [9].

With the aim of assisting European municipalities in planning and timing preventive vector control, we assessed the potential epidemiological impact on chikungunya and dengue, and the ensuing economic benefits for the health system, produced by routine larviciding against *Ae. albopictus* within urban sites in temperate climates.

Methods

Mosquito data

Mosquito monitoring via adult trapping was carried out in ten municipalities from the Northern Italian provinces of Belluno and Trento, characterized by a temperate climate [10]. Mosquitoes were collected using Biogents (BG) Sentinel traps (Biogents AG, Regensburg, Germany) baited with lures and CO_2 from dry ice. After each trapping session, mosquitoes were killed by freezing at -20°C, identified using taxonomic keys [11, 12] and confirmed by PCR if found in a location for the first time [12].

Transmission dynamics model

We simulated the transmission dynamics associated with chikungunya and dengue using a standard SEI-SEIR approach [13] in which mosquitoes develop lifelong infection after an (extrinsic) incubation period since the bite to an infectious human (SEI sub-model), whereas humans develop temporary infection, followed by the development of immunity, after an (intrinsic) incubation period since the bite from an infectious mosquito (SEIR sub-model). We considered temperature-dependent extrinsic incubation periods and per-bite transmission probabilities for dengue [14], whereas only temperature-independent estimates were available for Chikungunya [15, 16]. The transmission model was initialized with a single infectious human, representing an imported case at a date sampled uniformly between January 1st and December 31st. The population size of female Ae. albopictus mosquitoes over time in the transmission model was estimated by fitting a population model to capture data collected in the absence of larvicidal treatments, following the same approach already adopted in [13, 17]. The model considers four developmental stages of mosquitoes (eggs, larvae, pupae and adults) and reproduces their life cycle by means of temperature-dependent parameters regulating the stage-specific rates of mortality and development. Free model parameters (i.e. the site- and year- specific habitat suitability and the capture rate of BG traps) were estimated via a Monte Carlo Markov Chain approach based on a Poisson likelihood [13, 17].

We then included the effect of routine larviciding in the population model. Experimental studies of several available commercial larvicide products show that 99% of existing larvae and hatching eggs are killed within a given breeding site, with constant efficacy of about 30 days, independently of the specific product used [5, 18, 19]. We considered a standard approach targeting breeding sites in publicly accessible spaces (e.g., catch basins placed in public parks and along the road system), and an additional strategy where public interventions were integrated by the involvement of citizens in treating and removing breeding sites within private premises. The latter was parametrized on results from a pilot project conducted in two municipalities within the same area of this study [7], in which larvicide products were delivered door-to-door and free of charge to house dwellers, who were sensitized and educated to mosquito control interventions. A key determinant of effectiveness is the fraction of existing breeding sites in a given area that are actually treated (coverage). We adopted a coverage range between 30% and 50% for larviciding of public catch basins only, and between 60% and 75% for interventions that additionally involve citizens. These ranges were computed from available data on the density and proportion of reachable breeding sites in public and private premises [7]. Other strategies aimed at extending the coverage (e.g. removal of other breeding sites such as water buckets, plant saucers, tarpaulins, etc.) were not considered.

Since the effect of larvicides is transitory, treatment of catch basins may be repeated multiple times within a given season. We considered several different starting dates and from 1 to 4 applications of larvicide treatments within a given mosquito season (hereafter referred to as "effort level"), implemented with monthly frequency.

Economic assessment

To evaluate the economic acceptability of the two considered strategies, a cost-utility analysis for the prevention of dengue and chikungunya was conducted, taking the number of infections

as input from the transmission model. Disability Adjusted Life Years (DALYs) averted and net costs were derived comparing an intervention scenario to the case in which no control programs were put in place (baseline). The baseline was set to reflect a municipality where only the monitoring of mosquito presence via ovitraps was performed [7].

The analysis was conducted from a public healthcare system perspective through the maximization of the net health benefit (NHB) [20]. This measure is defined as the difference between the DALY averted and the incremental cost due to the intervention, the latter divided by the willingness to pay (WTP) by public authorities for each DALY averted. Following WHO recommendations [21], we assumed such value approximately equal to the Gross Domestic Product (GDP), which is about 35,000 euro per capita in our study area [22].

Probabilities of each infected case of being symptomatic, notified, severe, hospitalized and of dying, and the length of stay in hospital, were derived from published studies [23, 24] and from analyzing data from the Italian Hospital Discharge System (Schede di Dimissioni Ospedaliere), accounting for all hospital admissions for chikungunya and dengue recorded in Italy. The cost of illness was estimated according to expert opinion. The costs of intervention were estimated from actual costs during control activities against *Ae. albopictus* recently performed in two municipalities from the study area [7].

For all the considered scenarios, the NHB was computed on a set of 100,000 stochastic realizations accounting for the uncertainty in both the transmission and the economic model's parameters. Full details on this analysis are provided in <u>S1 Text</u>.

To assess the feasibility and sustainability of public interventions, we used responses from a questionnaire administered in 2013 to municipalities of the province of Trento, aimed at collecting information on the actual public expenditure on vector control activities.

Results

The estimated density of adult female mosquitoes (averaged between April 10th and September 30th) was between 4 and 88 per hectare in 2014 and between 9 and 198 in 2015, depending on the municipality (see <u>Table 1</u>). The higher abundance in 2015 is mostly due to the much higher

Municipality	Mosquito density (adult mosquitoes /		Population (inhabitants)	Urban surface (ha)	Population density (inhabitants / ha)	
	na)					
Strigno *	3.8	8.9	3354	85	39.5	
Belluno	9.8	27.1	35703	539	66.2	
Tenno	14.1	88.9	1823	23	79.3	
Tezze**	19.9	72.1	1742	36	48.4	
Povo	42.4	138.5	5571	88	63.3	
Santa Giustina	46.4	95.6	6800	159	42.8	
Trento	52.3	128.4	117304	1570	74.7	
Riva del Garda ***	72.7	197.5	32259	418	77.2	
Feltre	74.2	161.1	20560	329	62.5	
Rovereto	87.6	-	39099	444	88.1	

Table 1. Estimated mosquito densities (average between April 10th–September 30th) and other relevant characteristics of the considered municipalities.

*: includes the neighbouring municipality of Telve

**: includes the neighbouring municipality of Villa Agnedo

***: includes the neighbouring municipality of Arco

https://doi.org/10.1371/journal.pntd.0005918.t001

PLOS Neglected Tropical Diseases | https://doi.org/10.1371/journal.pntd.0005918 September 11, 2017

temperatures recorded during summer. The initial reproduction numbers and the threshold for autochthonous transmission of chikungunya and dengue over time were estimated in a previous study [17]. Here, for each site and year, we computed the probability of autochthonous transmission of chikungunya and dengue originated by an imported infection in the absence of interventions. Higher vector densities during 2015 resulted in an increased risk of local transmission for both infections, compared to the previous year. The probability of observing at least one secondary case was estimated to be up to 30% for chikungunya and 15% for dengue in highly infested towns in 2015. Corresponding maximum probabilities in 2014 were around 20% for chikungunya and 5% for dengue. This means that 7 importations of chikungunya and 5% for dengue. This means that 7 importations of chikungunya and 15 importations of dengue in towns most at risk would have a >90% probability of causing at least one secondary case in 2015. Sporadic transmission (less than 10 secondary cases) is by far the most likely scenario, especially for dengue (Fig 1). However, we found a low, but non-negligible, probability (up to 2.7%) that an uncontrolled chikungunya outbreak would produce more than 50 cases in several sites during 2015.

Routine preventive larvicide treatments can reduce significantly mosquito populations and consequently the probability and size of outbreaks triggered by sporadic importation of infected cases. To evaluate the overall effectiveness, we considered the expected number of total secondary infections per imported case. Under the baseline scenario of no control interventions, this index ranged from 0.1 to 5.2, depending on the site and year; corresponding numbers for dengue were everywhere below 0.5. Because of the smaller epidemiological risk of dengue, we discuss only the cost-effectiveness analysis on chikungunya, leaving corresponding results for dengue to the S1 Text.

For each site and year, and for each timing, effort level and assumed coverage, we evaluated the relative reduction in the expected number of secondary infections per imported case as a measure of effectiveness. Fig 2 and Table 2 show that all interventions with optimal effectiveness covered the month of July, which corresponds to the estimated period of steepest growth of the adult *Ae. albopictus* population in both years. We selected for further analyses only interventions with optimal timing for each effort level (Table 2; the reduction in mosquito abundance corresponding to the optimally timed interventions is reported in the <u>S1 Text</u>). We found that an increase in the effort level does not proportionally reduce the expected number of cases (Fig 3). In particular, an expansion in the coverage of breeding sites from 30% to 50% would be more effective than doubling the effort level while keeping the coverage at 30%. In general, interventions are most beneficial when the baseline risk is highest.

Towards an optimal allocation of resources, the benefits of reducing the potential number of transmitted cases needs to be compared with the intervention costs. Taking into account all possible clinical outcomes, including the probability of severe illness and of hospitalization, the estimated average cost per infection is 424.9 euros (95% CI 342-533) for chikungunya and 275.88 euros (95% CI 151-422) for each dengue infection. The corresponding average DALY loss per case is higher for chikungunya (0.45, 95% CI 0.10–1.12) than for dengue (0.29, 95% CI 0.15–0.44). In Fig 4, we show the relative probability that each effort level (including the nointervention scenario) will maximize the NHB for each site, year, and coverage. Three main outcomes can be identified. The first is represented by larger cities (Trento, Belluno and Rovereto, all above 35,000 inhabitants) where non-intervention has the highest likelihood of being optimal. In these sites, the poor economic effectiveness of larviciding depends on the relatively low number of expected secondary cases even in the absence of treatment (Fig 3), combined with the high intervention costs due to the extent of the area to be covered. The second group consists of smaller towns where intervention is always beneficial (Povo, Santa Giustina, Tenno and Tezze, all below 10,000 inhabitants) and where higher effort levels have the highest probabilities of being optimal. Strigno (about 3,400 inhabitants) represents an exception to



Fig 1. Epidemiological predictions in the absence of control interventions. Probability of local transmission of chikungunya (CHIKV) and dengue (DENV), disaggregated by outbreak size, in the 10 study locations during the mosquito seasons.



Fig 2. Effectiveness of larviciding in reducing the expected number of secondary CHIKV infections. Distribution across all sites and both years, disaggregated by coverage value (lighter colour: 30%; darker colour: 50%), intervention timing (starting date every 15 days between May 1st and September 1st) and effort level (i.e., number of larvicide applications with monthly frequency (from 1 to 4). Grey bars: optimal starting date; shaded area: duration of intervention.

https://doi.org/10.1371/journal.pntd.0005918.g002

this rule, where the low intervention costs are counterbalanced by a very small transmission risk in the absence of interventions. Nonetheless, even in Strigno a low-effort intervention (single treatment) might be beneficial because of its low cost. The third situation occurs in towns of intermediate size (Feltre and Riva del Garda, between 20,000 and 35,000 inhabitants) where both the intervention costs and the transmission risks are high. In these cases, depending on the larviciding coverage, absence of intervention might be the optimal strategy in seasons of lower mosquito abundance (2014) while a low-to-moderate effort (1 to 3 treatments) might be the best choice in years of high infestation (2015). Overall, the probability that a more intensive intervention will be optimal increases with the coverage and with higher transmission risk (2015 vs. 2014). We also tested the cost-effectiveness of expanding the coverage

Table 2. Timing of optimal intervention and estimated reduction in the number of chikungunya cases at 30% and 50% coverage (mean and range across all sites) by effort level (one to four larvicide applications over a season).

Effort level	Optimal starting date	End of efficacy, optimal timing	Reduction in number of CHIKV cases, % (range)		
			30% coverage	50% coverage	
1 intervention	July 1 st	August 1 st	18.6 (0–36.4) %	33.3 (13.9–48.9) %	
2 interventions	June 1 st	August 1 st	32.3 (6.9–49.8) %	51.2 (27.7–66.6) %	
3 interventions	May 15 th	August 15 th	40.5 (20.8–54.4) %	61.5 (39.0–74.0) %	
4 interventions	May 1 st	September 1 st	43.1 (23.7–57.3) %	64.9 (39.2–77.2) %	



Fig 3. Estimated effectiveness of optimally timed interventions for different coverages (30% and 50%) and effort level (0–4), disaggregated by site and year.





by involving private citizens [7]. We found that this type of intervention might achieve significant additional reductions in the expected number of secondary cases and probability of local transmission (reported in the S1 Text). However, they are rarely optimal from the economic perspective because they require labour-intensive activities. Fig.5 reports results of the NHB analysis for a single larvicide treatment, but qualitative inferences are similar for more intensive efforts (see S1 Text). The only two instances where involvement of citizens was found to be economically beneficial were Povo and Tezze and only during the 2015 mosquito season, i.e. only where the urban size is small enough to keep intervention costs low and where the transmission risk at baseline is sufficiently high.

Two municipalities under study, Trento and Riva del Garda, had responded to a previously administered questionnaire on public expenditure on vector control, declaring an overall budget of 0.254 euro and 0.532 euros per inhabitant, respectively. In Trento, the most cost-effective activity predicted by our model was monitoring by ovitraps (Fig 4), which has an estimated average cost of 0.016 euro per inhabitant; in Riva del Garda, one or two larvicide applications per year would be likely optimal and would cost between 0.256 and 0.512 euros per inhabitant. Therefore, the most cost-effective strategies are sustainable with respect to the current allocated budget. We provide full details on questionnaires, municipality-specific answers and intervention costs in the S1 Text.

Discussion

In this work, we evaluated the effect of routine larviciding against dengue and chikungunya, two viruses transmitted by bites of *Ae. albopictus* mosquitoes. We used data from two seasons of entomological surveillance in multiple sites from northern Italy to parametrize a mathematical model of mosquito population dynamics and control. The population model was coupled with a transmission dynamics model and a cost-effectiveness analysis to identify suitable routine vector control strategies for temperate climate municipalities in Europe. We found that, in the absence of interventions, the risk of autochthonous dengue transmission was low and limited to sporadic transmission in both years, because of the relatively low competence of European strains of *Ae. albopictus*. On the other hand, the risk of a chikungunya outbreak was estimated to be up to 30% in 2015, with a non-negligible probability of observing outbreaks larger than 50 cases in most sites.

We found that the most effective interventions in reducing the amount of expected locally transmitted cases were those for which the window of larvicide efficacy covered at least the month of July (Fig 2, Table 2). Larviciding reduced the probability of secondary cases only moderately, but it had an important impact in avoiding larger outbreaks. Our analysis included two seasons that were representative of a broad range of mosquito abundances, due to the remarkable temperature differences. The cost-effectiveness of larviciding depends on the actual mosquito abundance in a given year; however, general rules could be identified independently of the considered year: small villages (<10,000 inhabitants) with moderate-tohigh mosquito abundances will maximally benefit of intense larviciding efforts made of season-round monthly treatment of public catch basins. For medium-sized towns (20-35,000 inhabitants) with high infestation rate, the benefits are partially offset by the higher cost of intervention; in these cases, a moderate larviciding effort (1 to 3 treatments within the season) is recommended. Larger cities in our study (>35,000 inhabitants) were characterized by a low or intermediate transmission risk, and the high costs of an intervention covering the entire urban area made it economically disadvantageous. In these situations, treating specific neighbourhoods with highest mosquito abundance (called 'hot spot' approach [25]) may be costeffective. In order to evaluate such a scenario, however, it would be necessary to model the



Fig 5. Probability of highest net health benefit for a single larvicide application according to the type of intervention (none vs. public vs. public and private), by year, coverage (30 and 50% for public intervention; 60 and 75% for both public and private intervention) and study site.

complex effect of the urban layout on the spatial distribution of breeding sites and on the dynamics of mosquito populations [7], which is out of the scope of our study. Treatment of private breeding sites via the direct involvement of citizens by door-to-door visits was recommended only in small towns with high mosquito infestation. A survey on the allocated budget for mosquito control programs across different municipalities showed that expenses required for the most cost-effective interventions are sustainable for the considered area.

These results need to be contextualized with respect to our simplifying assumptions. First, all results are given conditionally on a uniform probability of importation of an infectious individual within a given epidemiological year. For comparison, in the considered provinces of Trento and Belluno, three imported cases of dengue and one imported case of chikungunya were recorded in 2014 (C. Rizzo, personal communication); however, the actual importation rate may vary significantly by year and time of the year, depending on spatio-temporal patterns of global epidemics and international travel. We did not consider reactive interventions that are implemented when a case of chikungunya or dengue infection is detected or after an outbreak has started (e.g., insecticide air spraying in the neighbourhood of the index case [26]). In addition, our results are relative to the prevention of arboviral transmission; however, there may be other purposes in vector control activities, such as the reduction of nuisance for citizens, which were not included in our analysis. For what concerns the economic assessment, we did not consider the impact of local transmission detection on the blood supply chain. Upon clinical confirmation of a locally transmitted arboviral infection, restrictions on the usage of blood bags collected in the region are applied to prevent transmission via transfusions, and screening tests on available blood supplies are implemented [26]. These additional interventions are quite expensive, and savings associated to the reduction of transmission risk granted by larvicides may dramatically offset the cost-benefit balance in favour of the intervention. However, these costs are difficult to estimate because of the lack of sufficient data.

We did not include other arboviroses transmitted by *Ae. albopictus* because of their lower epidemiological relevance to the considered area. For example, the risk of Zika virus transmission was found to be close to zero in the study region, even under conservative scenarios [17]. Nonetheless, we note that larvicides produce simultaneous benefits in preventing multiple diseases transmitted not only by *Ae. albopictus* but also by other affected mosquito species (e.g. West Nile virus associated to *Culex pipiens* L.). Furthermore, larviciding may assist in limiting the spread of other invasive mosquito species such as *Aedes (Hulecoeteomyia) japonicus* (Theobald) and *Aedes (Hulecoeteomyia) koreicus* (Edwards) [1, 27]. An interesting research question is how the balance of ecological interactions between mosquito species [28] may be offset by such interventions.

Other studies [2, 6, 7] have investigated the effectiveness of vector control in Europe using different approaches. The cost-effectiveness of larvicidal treatment against *Ae. albopictus* in temperate climates has been evaluated only in combination with other interventions during an ongoing outbreak [29, 30]; other studies were based on endemic (extra-European) settings where transmission is mainly mediated by *Aedes (Stegomyia) aegypti* (Linneus) [31, 32]. Overall, results from different studies and approaches, including our own, are consistent in highlighting the potential of larviciding towards reducing mosquito populations; however, this reduction will not result in a complete elimination of the risk of local chikungunya or dengue transmission. Additional strategies may integrate the control of risks from mosquito-borne diseases, including source reduction methods (e.g. identification and removal of breeding sites), mass trapping (e.g. via lethal ovitraps) and approaches leveraging ecological interactions (such as the use of *Wolbachia* bacteria or the release of genetically sterilized male mosquitoes). A comprehensive review of the potential for these strategies can be found in [9], but specific cost-effectiveness studies are needed to identify optimal strategies for vector control. European

municipalities with temperate climate where *Ae. albopictus* is established may take advantage of results from this study when planning and timing routine larviciding interventions aimed to prevent or reduce epidemiological risks. Temperate European areas share with our study collection area similar temperature suitability for the transmission of arboviruses [33] and similar abundances of *Ae. albopictus* [34], so that results on the epidemiological effectiveness of larviciding should not differ significantly. More caution should be paid when extrapolating cost-effectiveness conclusions to different countries, given potential differences in health and intervention costs and in the choice of the WTP. Finally, we suggest that the proposed methodological approach may also be extended to European areas with different climates, conditional on the availability of local data on mosquito abundances estimated via entomological surveillance activities.

Supporting information

S1 Text. Technical details on the implementation of the transmission dynamics model, the cost-effectiveness analysis and additional model results. (PDF)

S1 Dataset. Capture data and daily temperatures for the 10 municipalities. (CSV)

Acknowledgments

We wish to thank Dr. Stefano Braschi and the Infectious Diseases Department at the San Matteo Hospital in Pavia (Italy) for providing data on the hospitalization costs of dengue and chikungunya. We also thank Caterina Rizzo for provision of data on chikungunya and dengue hospitalizations and importation.

Author Contributions

Conceptualization: Giorgio Guzzetta, Filippo Trentini, Piero Poletti, Stefano Merler, Alessia Melegaro.

Data curation: Giorgio Guzzetta, Filippo Trentini, Frederic Alexandre Baldacchino, Fabrizio Montarsi.

Funding acquisition: Gioia Capelli, Annapaola Rizzoli, Roberto Rosà, Stefano Merler, Alessia Melegaro.

Investigation: Giorgio Guzzetta.

Project administration: Annapaola Rizzoli.

Resources: Stefano Merler, Alessia Melegaro.

Software: Giorgio Guzzetta.

Supervision: Gioia Capelli, Roberto Rosà, Stefano Merler, Alessia Melegaro.

Visualization: Giorgio Guzzetta.

Writing - original draft: Giorgio Guzzetta, Filippo Trentini.

Writing – review & editing: Giorgio Guzzetta, Filippo Trentini, Piero Poletti, Frederic Alexandre Baldacchino, Fabrizio Montarsi, Gioia Capelli, Annapaola Rizzoli, Roberto Rosà, Stefano Merler, Alessia Melegaro.

References

- Medlock JM, Hansford KM, Schaffner F, Versteirt V, Hendrickx G, Zeller H, Van Bortel W. A review of the invasive mosquitoes in Europe: ecology, public health risks, and control options. Vector Borne Zoonotic Dis. 2012 12(6):435–47. https://doi.org/10.1089/vbz.2011.0814 PMID: 22448724
- Caputo B, Manica M, D'Alessandro A, Bottà G, Filipponi F, Protano C, Vitali M, Rosà R, della Torre A. Assessment of the Effectiveness of a Seasonal-Long Insecticide-Based Control Strategy against Aedes albopictus Nuisance in an Urban Area. PLoS Negl Trop Dis. 2016; 10(3):e0004463. https://doi. org/10.1371/journal.pntd.0004463 PMID: 26937958
- Burattini MN, Chen M, Chow A, Coutinho FA, Goh KT, Lopez LF, Ma S, Massad E. Modelling the control strategies against dengue in Singapore. Epidemiol Infect. 2008; 136(3):309–19. <u>https://doi.org/10. 1017/S0950268807008667</u> PMID: 17540051
- Sochacki T, Jourdain F, Perrin Y, Noel H, Paty MC, de Valk H, Septfons A, Simard F, Fontenille D, Roche B. Imported chikungunya cases in an area newly colonised by Aedes albopictus: mathematical assessment of the best public health strategy. Euro Surveill. 2016; 21(18).
- Bellini R, Albieri A, Carrieri M, Colonna R, Donati L, Magnani M, Pilani R, Veronesi R, Chiot G, Lanza N. Efficacy and lasting activity of four IGRs formulations against mosquitoes in catch basins of northern Italy. Eur Mosq Bull. 2009; 27:33–46.
- Abramides GC, Roiz D, Guitart R, Quintana S, Guerrero I, Giménez N. Effectiveness of a multiple intervention strategy for the control of the tiger mosquito (Aedes albopictus) in Spain. Transactions of the Royal Society of Tropical Medicine and Hygiene. 2011; 105(5):281–8. https://doi.org/10.1016/j.trstmh. 2011.01.003 PMID: 21466887
- Baldacchino F, Bussola F, Arnoldi D, Marcantonio M, Montarsi F, Capelli G, Rosà R, Rizzoli A. An integrated pest control strategy against the Asian tiger mosquito in northern Italy: a case study. Pest Management Science. 2017; 73(1):87–93. https://doi.org/10.1002/ps.4417 PMID: 27539880
- Luz PM, Struchiner CJ, Galvani AP. Modeling transmission dynamics and control of vector-borne neglected tropical diseases. PLoS Negl Trop Dis. 2010; 4(10):e761. <u>https://doi.org/10.1371/journal.pntd.0000761</u> PMID: 21049062
- Baldacchino F, Caputo B, Chandre F, Drago A, della Torre A, Montarsi F, Rizzoli A. Control methods against invasive Aedes mosquitoes in Europe: a review. Pest Manag Sci. 2015; 71(11):1471–85. <u>https://doi.org/10.1002/ps.4044</u> PMID: 26037532
- 10. Peel MC, Finlayson BL, McMahon TA. Updated world map of the Köppen-Geiger climate classification. Hydrol Earth Syst Sci. 2007; 11(5):1633–44. http://dx.doi.org/10.5194/hess-11-1633-2007
- 11. Becker N, Petric D, Zgomba M, Boase C, Madon M, Dahl C, Kaiser A. (2010) Mosquitoes and Their Control. 2nd edition. Heidelberg, Springer.
- Montarsi F, Martini S, Dal Pont M, Delai N, Milone NF, Mazzucato M, Soppelsa F, Cazzola L, Cazzin S, Ravagnan S, Ciocchetta S, Russo F, Capelli G. Distribution and habitat characterization of the recently introduced invasive mosquito Aedes koreicus (Hulecoeteomyia koreica), a new potential vector and pest in north-eastern Italy. Parasites & Vectors. 2013; 6, 292.
- Guzzetta G, Montarsi F, Baldacchino F, Metz M, Capelli G, Rizzoli A, Pugliese A, Rosà R, Poletti P, Merler S. Potential risk of dengue and chikungunya outbreaks in northern Italy based on a population model of Aedes albopictus (Diptera: Culicidae). PLos Negl Trop Dis. 2016.
- 14. Focks D, Daniels E, Haile DG, Keesling JE. A simulation model of the epidemiology of urban dengue fever: literature analysis, model development, preliminary validation, and samples of simulation results. The American journal of tropical medicine and hygiene. 1995; 53: 489–506 PMID: 7485707
- Poletti P, Messeri G, Ajelli M, Vallorani R, Rizzo C, Merler S. Transmission potential of chikungunya virus and control measures: the case of Italy. PLoS One. 2011; 6(5):e18860. <u>https://doi.org/10.1371/</u> journal.pone.0018860 PMID: 21559329
- Waldock J, Chandra NL, Lelieveld J, Proestos Y, Michael E, Christophides G, Parham PE. The role of environmental variables on Aedes albopictus biology and chikungunya epidemiology. Pathogens and global health. 2013; 107(5): 224–241. <u>https://doi.org/10.1179/2047773213Y.0000000100</u> PMID: 23916332
- 17. Guzzetta G, Poletti P, Montarsi F, Baldacchino F, Capelli G, Rizzoli A, Rosà R, Merler S. Assessing the potential risk of Zika virus epidemics in temperate areas with established Aedes albopictus populations. Euro surveillance. 2016; 21(15).
- Xue RD, Barnard DR, Ali A. Laboratory evaluation of 21 insect repellents as larvicides and as oviposition deterrents of Aedes albopictus (Diptera: Culicidae). Journal of the American Mosquito Control Association. 2006; 22(1):126. <u>https://doi.org/10.2987/8756-971X(2006)22[126:LEOIRA]2.0.CO;2</u> PMID: 16646335

- Carrieri M, Masetti A, Albieri A, Maccagnani B, Bellini R. Larvicidal activity and influence of Bacillus thuringiensis var. israelensis on Aedes albopictus oviposition in ovitraps during a two-week check interval protocol. Journal of the American Mosquito Control Association. 2009; 25(2): 149–155. <u>https://doi.org/ 10.2987/08-5852.1</u> PMID: 19653496
- 20. Stinnet A., Mullay J. Net Health Benefits: A New Framework for the Analysis of Uncertainty in Cost-Effectiveness Analysis. NBER Technical Working Paper. 1998;227
- World Health Organization (2002). The world health report 2002: reducing risks, promoting healthy life. Geneva.
- 22. Statistiche Istat. http://dati.istat.it/Index.aspx; accessed on Dec 27, 2016.
- 23. World Health Organization (2004). Disability weights for diseases and conditions. In Global Burden of Disease 2004 update. Geneva.
- LaBeaud AD, Bashir F, King CH. Measuring the burden of arboviral diseases: the spectrum of morbidity and mortality from four prevalent infections. Population Health Metrics. 2011; 9:1. https://doi.org/10. 1186/1478-7954-9-1 PMID: 21219615
- Unlu I, Klingler K, Indelicato N, Faraji A, Strickman D. Suppression of Aedes albopictus, the Asian tiger mosquito, using a 'hot spot' approach. Pest Manag Sci. 2016 Jul; 72(7):1427–32. <u>https://doi.org/10.1002/ps.4174 PMID: 26482455</u>
- 26. Ministero della Salute. Sorveglianza dei casi umani delle malattie trasmesse da vettori con particolare riferimento a Chikungunya, Dengue, Zika virus e West Nile Disease–2016.
- 27. Capelli G, Drago A, Martini S, Montarsi F, Soppelsa M, Delai N, Ravagnan S, Mazzon L, Schaffner F, Mathis A, Di Luca M. First report in Italy of the exotic mosquito species Aedes (Finlaya) koreicus, a potential vector of arboviruses and filariae. Parasites & vectors. 2011; 4(1):1.
- Marini G, Guzzetta G, Rosà R, Merler S. The effect of interspecific competition on the temporal dynamics of Aedes albopictus and Culex pipiens. Parasites and Vectors, 2017; in press.
- Poletti P, Messeri G, Ajelli M, Vallorani R, Rizzo C, Merler S. Transmission potential of chikungunya virus and control measures: the case of Italy. PLoS One. 2011; 6(5):e18860. https://doi.org/10.1371/ journal.pone.0018860 PMID: 21559329
- Mecoli M, De Angelis V, Brailsford SC. Using system dynamics to evaluate control strategies for mosquito-borne diseases spread by human travel. Computers & Operations Research. 2013; 40(9):2219– 28.
- 31. Tan AW, Loke SR, Benjamin S, Lee HL, Chooi KH, Sofian-Azirun M. Spray application of Bacillus thuringiensis israelensis (Bti strain AM65-52) against Aedes aegypti (L.) and Ae. albopictus Skuse populations and impact on dengue transmission in a dengue endemic residential site in Malaysia. Southeast Asian J Trop Med Public Health. 2012 Mar; 43(2):296–310. PMID: 23082582
- 32. Tan AW, Loke SR, Benjamin S, Lee HL, Chooi KH, Sofian-Azirun M. Spray application of Bacillus thuringiensis israelensis (Bti strain AM65-52) against Aedes aegypti (L.) and Ae. albopictus Skuse populations and impact on dengue transmission in a dengue endemic residential site in Malaysia. Southeast Asian J Trop Med Public Health. 2012 Mar; 43(2):296–310. PMID: 23082582
- Brady OJ, Golding N, Pigott DM, Kraemer MU, Messina JP, Reiner RC, Hay SI. Global temperature constraints on Aedes aegypti and Ae. albopictus persistence and competence for dengue virus transmission. Parasites & vectors. 2014; 7(1): 338.
- 34. Kraemer MU, Sinka ME, Duda KA, Mylne A, Shearer FM, Brady OJ, Messina JP, Barker CM, Moore CG, Carvalho RG, Coelho GE, Van Bortel W, Hendrickx G, Schaffner F, Wint GR, Elyazar IR, Teng HJ, Hay SI. The global compendium of *Aedes aegypti* and *Ae. albopictus* occurrence. Sci Data. 2015; 2:150035. https://doi.org/10.1038/sdata.2015.35 PMID: 26175912

Reproduced with permission of copyright owner. Further reproduction prohibited without permission.



www.manaraa.com